Author's response to reviews

Title: Council Tax Valuation Band predicts breast feeding and socio-economic status in the Alspac study population.

Authors:

Norman Beale (info@northlandsrnd.org.uk)
Gill Kane (info@northlandsrnd.org.uk)
Mark Gwynne (info@northlandsrnd.org.uk)
Carole Peart (info@northlandsrnd.org.uk)
Gordon Taylor (g.j.taylor@bath.ac.uk)
David Herrick (david.herrick@bristol.ac.uk)
Andy Boyd (a.w.boyd@bristol.ac.uk)

Version: 3 Date: 23 November 2005

Author’s response to reviews:

Dear Editors - have changed the manuscript according to your requests of 15 November viz. the format of the affiliations of author; renaming introduction and methods; eliminating highlighted previous changes. You should note that I have been unable to open the new comments of the referee - the tables (a) to (j) that she requested are still at the end of the manuscript but I'm unsure if they need to be published in the paper - they were requested as supplementary evidence. Norman Beale.